Author's response to reviews

Title: Genome Screen in Familial Intracranial Aneurysm

Authors:

Tatiana Foroud (tforoud@iupui.edu)
Laura Sauerbeck (SAUERBLR@ucmail.uc.edu)
Robert Brown (brown@mayo.edu)
Craig Anderson (canderson@george.org.au)
Daniel Woo (WOODL@ucmail.uc.edu)
Dawn Kleindorfer (KLEINDDO@ucmail.uc.edu)
Matthew L. Flaherty (matthew.flaherty@uc.edu)
Ranjan Deka (DEKAR@ucmail.uc.edu)
Richard Hornung (richard.hornung@cchmc.org)
Irene Meissner (meissner.irene@mayo.edu)
Joan E. Bailey-Wilson (jebw@nhgri.nih.gov)
Carl Langefeld (clangefe@wfubmc.edu)
Guy Rouleau (guy.rouleau@umontreal.ca)
E. Sander Connolly (esc5@columbia.edu)
Dongbing Lai (dlai@iupui.edu)
Daniel L. Koller (dkoller@iupui.edu)
John Huston III (jhuston@mayo.edu)
Joseph P. Broderick (broderjp@ucmail.edu)
FIA Study Investigators [Familial Intracranial Aneurysm Study Investigators] (tforoud@iupui.edu)

Version: 7 Date: 12 January 2009

Author's response to reviews:

I realized that there was still a tracked page break in the document, so I have accepted that tracked change and am reuploading